References for Analysis of Local Growth Models These are not meant to be exhaustive, but rather a few pointers.

Lecture 1. Shapes in deterministic and random rules.

J. Gravner, D. Griffeath, Cellular automaton growth on Z^2: theorems, examples and problems. Advances in Applied Mathematics 21 (1998), 241–304.

J. Gravner, D. Griffeath, Nucleation parameters in discrete threshold growth dynamics. Experimental Math. 6 (1997), 207–220.

T. Bohman, J. Gravner, Random Threshold Growth dynamics. Random Structures and Algorithms 15 (1999), 93–111.

J. M. Steele, “Probability Theory and Combinatorial Optimization.” SIAM, 1997. J. T. Cox, R. Durrett, Limit theorems for the spread of epidemics and forest fires. Stochastic Process. Appl. 30 (1988), 171–191.

Lecture 2. Toom’s methods and applications.

A. L. Toom, Nonergodic multidimensional systems of automata. Problemy Peredači Informacii 10 (1974), 70–79.

R. Durrett, T. M. Liggett, The shape of the limit set in Richardson’s growth model. Ann. Probab. 9 (1981), 186–193.

J. L. Lebowitz, C. Maes, E. R. Speer, Statistical mechanics of probabilistic cellular automata. J. Stat. Phys. 59 (1990), 117–168.

M. Bramson, L. Gray, A useful renormalization argument. In “Random walks, Brownian motion, and interacting particle systems. Festshrift in Honor of Frank Spitzer,” R. Durrett and H. Kesten, eds., Birkhüser, 1991. Pgs. 113–152.

A. L. Toom, Cellular automata with errors: problems for students of probability. In “Topics 1

2 in contemporary probability and its applications, ” J. L. Snell, ed., CRC Press, 1995. Pgs. 117–157.

Lecture 3. An exactly solvable growth model.

H. Kesten, R. H. Schonmann, On some growth models with a small parameter, Probab. Th. Rel. Fields 101 (1995), 435–468.

J. Gravner, Recurrent ring dynamics in two–dimensional excitable cellular automata. J. Appl. Prob. 36 (1999), 492–511.

J. Gravner, C. A. Tracy, H. Widom, Limit theorems for height fluctuations in a class of discrete space and time growth models. J. Statist. Phys. 102 (2001), 1085–1132. M. Prähofer, H. Spohn, Statistical self-similarity of one-dimensional growth processes. Phys. A 279 (2000), 342–352.

T. Seppäläinen, Exact limiting shape for a simplified model of first–passage percolation on the plane. Ann. Prob. 26 (1998), 1232–1250.

Lecture 4. Critical growth in random environment.

M. Aizenman, J. L. Lebowitz, Metastability effects in bootstrap percolation, J. Phys. A 21 (1988), 3801–3813.

A. Holroyd, Sharp metastability threshold for two–dimensional bootstrap percolation, preprint. J. Gravner, E. McDonald, Bootstrap percolation in a polluted environment. J. Stat. Phys. 87 (1997), 915–927.